Author's response to reviews

Title: Comparative Chest Computed Tomography Findings of Non-tuberculous Mycobacterial Lung Diseases and Pulmonary Tuberculosis in Patients with AFB Smear-Positive Sputum

Authors:

Yuan Mei-Kang (12703@ymuh.ym.edu.tw)
Chang Cheng Yu (koala2716@hotmail.com)
Chang Shih-Chieh (11319@ymuh.ym.edu.tw)
Tsai Ping-Huang (badaucheng@yahoo.com.tw)
Lee Yuan-Ming (koala2716@gamil.com)
Huang Jen-Wu (dtsurga9@yahoo.com.tw)

Version: 7
Date: 10 April 2014

Author's response to reviews:

We had modified the format and submitted revised version of the manuscript already.

1. Title page:
We had modified the title page including the email addresses of all authors.

2. Copyedit:
All participants were gave written informed consent and agree of information collection in the study. We also had modified the sentence in page 7 also.

3. Figures:
We have reviewed all the figures and ascertained that these CT images were extracted from standard DICOM export of PACS (Matrix 512x512 for CT). We transformed these selected images into TIF. with 300 dpi for the use of standard academic publication.